Author's response to reviews

Title: Functions of behavior change interventions when implementing multi-professional teamwork at an emergency department: a comparative case study

Authors:

Mandus Frykman (mandus.frykman@ki.se)
Henna Hasson (henna.hasson@ki.se)
Åsa Muntlin Athlin (asa.muntlin@pubcare.uu.se)
Ulrica von Thiele Schwarz (ulrica.schwarz@ki.se)

Version: 3 Date: 14 April 2014

Author's response to reviews: see over
14th April 2014

Functions of behavior change interventions when implementing teamwork at an emergency department: a comparative case study

Dear Editor,

Thank you for recommending our manuscript for publication. As requested the name of the ethical committee that approved this study is added on page 12.

Sincerely yours

Mandus Frykman, MSc, PhD Candidate
LIME, Dept. of Learning, Informatics, Management and Ethics
Karolinska Institutet
S- 171 77 Stockholm
+46 737 24 91 21
mandus.frykman@ki.se